\documentclass[12pt,reqno]{article}

\usepackage[usenames]{color}
\usepackage{amssymb}
\usepackage{graphicx}
\usepackage{amscd}

\usepackage[colorlinks=true,
linkcolor=webgreen,
filecolor=webbrown,
citecolor=webgreen]{hyperref}

\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}

\usepackage{color}
\usepackage{fullpage}
\usepackage{float}

\usepackage{psfig}
\usepackage{graphics,amsmath,amssymb}
\usepackage{amsthm}
\usepackage{amsfonts}
\usepackage{latexsym}
\usepackage{epsf}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{.1in}
\setlength{\evensidemargin}{.1in}
\setlength{\topmargin}{-.5in}
\setlength{\textheight}{8.9in}

\newcommand{\seqnum}[1]{\href{http://oeis.org/#1}{\underline{#1}}}

\begin{document}
\begin{thebibliography}{99}
\bibitem{0.01} V. S. Adamchik, On Stirling numbers and Euler sums, \textit{%
J. Comput. Appl. Math. }\textbf{79} (1997), 119--130.

\bibitem{Choi} J. Choi, Certain summation formulas involving harmonic
numbers and generalized harmonic numbers, \textit{Appl. Math. Comput. }%
\textbf{218} (2011), 734--740.

\bibitem{ChoiSriv} J. Choi and H. M. Srivastava, Some summation formulas
involving harmonic numbers and generalized harmonic numbers, \textit{Math.
Comp. Modelling }\textbf{54 }(2011), 2220--2234.

\bibitem{02} W. Chu, Summation formulae involving harmonic numbers, \textit{%
Filomat} \textbf{26} (2012), 143--152.

\bibitem{ChuYan} W. Chu and Q. Yan, Combinatorial identities on binomial
coefficients and harmonic numbers, \textit{Util. Math. }\textbf{75 }(2008),
51--66.

\bibitem{2.1b} P. Flajolet and R. Sedgewick, Mellin transforms and
asymptotics: finite differences and Rice's integrals, \textit{Theoret.
Comput. Sci. }\textbf{144} (1995), 101--124.

\bibitem{2.2} R. L. Graham, D. E. Knuth, and O. Patashnik, \textit{Concrete
Mathematics}, Addison-Wesley, 2nd Ed., 1994.

\bibitem{10b} M. Petkov\v{s}ek, H. S. Wilf, and D. Zeilberger, $A=B$, A. K.
Peters, , 1996.

\bibitem{11b} A. Sofo, \textit{Computational Techniques For The Summation Of
Series}, Kluwer Academic/Plenum Publishers, 2003.

\bibitem{12b} A. Sofo, Summation formula involving harmonic numbers, \textit{%
Anal. Mathematica} \textbf{37} (2011), 51--64.

\bibitem{13b} A. Sofo, Harmonic sums and integral representations, \textit{%
J. Appl. Analysis }\textbf{16} (2010), 265--277.

\bibitem{13c} A. Sofo, Integral forms of sums associated with harmonic
numbers, \textit{Appl. Math. Comput. }\textbf{207 }(2009), 365--372.

\bibitem{13d} A. Sofo, Sums of derivatives of binomial coefficients, \textit{%
Advances Appl. Math.} \textbf{42 }(2009), 123--134.

\bibitem{sofsriv} A. Sofo and H. M. Srivastava, Identities for the harmonic
numbers and binomial coefficients, \textit{Ramanujan J. }\textbf{25 }(2011),
93--113.

\bibitem{SunZhao} Z. W. Sun and L. L. Zhao, Arithmetic theory of harmonic
numbers (II), \newline
\url{http://arxiv.org/arXiv:0911.4433V7}.

\bibitem{14a} J. Spie\ss , Some identities involving harmonic numbers.
\textit{Math. Comput.} \textbf{55 }(1990), 839--863.
\end{thebibliography}


\begin{center}
\epsfxsize=4in
\leavevmode\epsffile{logo129.eps}
\end{center}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}

\begin{center}
\vskip 1cm{\LARGE\bf New Classes of Harmonic Number Identities
}
\vskip 1cm
\large
Anthony Sofo \\
Applied Mathematics \\
Victoria University \\
Melbourne, Victoria 8001\\
Australia\\
\href{mailto:anthony.sofo@vu.edu.au}{\tt anthony.sofo@vu.edu.au} \\
\end{center}

\vskip .2 in

\def\func#1{\mathop{\rm #1}\nolimits}%

\begin{abstract}
We develop some new classes of harmonic number identities, and
give an integral proof of an identity given by Sun and Zhao.
\end{abstract}

\section{Introduction and preliminaries}

In the paper \cite{SunZhao}, Sun and Zhao quote the following lemma.

\begin{lemma}
\label{lemma01} The following identity holds
\begin{equation}
\sum\limits_{n=1}^{p}\binom{p}{n}\frac{\left( -1\right) ^{n+1}~H_{n}}{n}%
=H_{p}^{\left( 2\right) }~\left( p\in \mathbb{N}\right) ,  \label{1.01}
\end{equation}%
where $\mathbb{N}:=\left\{ 1,2,3,\ldots \right\} .$
\end{lemma}

They state that (\ref{1.01}) is known, but it is difficult to give a
reference and then proceed to give a proof by induction. The induction proof
is perfectly valid however it does not shed light on how and if (\ref{1.01})
may be generalized. In this paper we give an integral representation for the
left hand side of (\ref{1.01}) and then generalize the result in various
directions. Some usual notational terms are now defined and will be utilized
throughout this paper. The generalized hypergeometric function notation
\begin{equation}
_{p}F_{q}\left[ \left.
\begin{array}{c}
a_{1},\dots ,a_{p} \\
b_{1},\dots ,b_{q}%
\end{array}%
\right\vert t\right] =\sum\limits_{r=0}^{\infty }\frac{\left( a_{1}\right)
_{r}\dots \left( a_{p}\right) _{r}}{r!\left( b_{1}\right) _{r}\dots \left(
b_{q}\right) _{r}}t^{r},\ \ \ p,~q\in
\mathbb{N}
_{0}:=
\mathbb{N}
\cup \left\{ 0\right\}\label{1.02}
\end{equation}%
where convergence is assured for
\begin{gather}
~p\leq q+1;~p\leq q~\text{and }\left\vert t\right\vert <\infty ; \nonumber\\
~p=q+1~\text{and }\left\vert t\right\vert <1;~ \nonumber\\
p=q+1,~\left\vert t\right\vert =1\text{ and }\func{Re}\left\{
\sum\limits_{m=1}^{q}b_{m}-\sum\limits_{m=1}^{p}a_{m}\right\}
>0,~b_{m},a_{m}\notin \left\{ 0,-1,-2,-3,\ldots \right\},\nonumber
\end{gather}
and $\left( a\right) _{r}$ is Pochhammer's symbol defined by $\left(
a\right) _{r}=a\left( a+1\right) \left( a+2\right) \cdots \left(
a+r-1\right) ,$ $r\in \mathbb{N},~\left( a\right) _{0}=1.$ The beta function
\begin{equation*}
B\left( s,t\right) =\int_{0}^{1}z^{s-1}\left( 1-z\right) ^{t-1}dz=\frac{%
\Gamma \left( s\right) \Gamma \left( t\right) }{\Gamma \left( s+t\right) }%
\text{ }\left( \func{Re}\left( s\right) >0\text{; }\func{Re}\left( t\right)
>0\right) ,
\end{equation*}%
and the gamma function
\begin{equation*}
\Gamma \left( z\right) =\int_{0}^{\infty }t^{z-1}e^{-t}dt~~\text{ }\left(
\func{Re}\left( z\right) >0\right) .
\end{equation*}%
The Riemann zeta function
\begin{equation*}
\zeta \left( z\right) =\sum_{r=1}^{\infty }\frac{1}{r^{z}}~~~\left( \func{Re}%
\left( z\right) >1\right)
\end{equation*}%
and the generalized harmonic numbers in power $\alpha $ are defined as
\begin{equation*}
H_{n}^{\left( \alpha \right) }=\sum\limits_{r=1}^{n}\frac{1}{r^{\alpha }}.
\end{equation*}%
The $n^{\text{th}}$ harmonic number, for $\alpha =1,$
\begin{equation*}
H_{n}^{\left( 1\right) }=H_{n}=\int_{0}^{1}\frac{1-t^{n}}{1-t}%
dt=\sum\limits_{r=1}^{n}\frac{1}{r}=\gamma +\psi \left( n+1\right) ,
\end{equation*}%
where $\gamma $ denotes the Euler-Mascheroni constant, defined by
\begin{equation*}
\gamma =\lim_{n\rightarrow \infty }\left( \sum\limits_{r=1}^{n}\frac{1}{r}%
-\log \left( n\right) \right) =-\psi \left( 1\right) \doteq
0.5772156649\dots
\end{equation*}%
The polygamma functions $\psi ^{\left( k\right) }\left( z\right) ,$ are
defined by%
\begin{align}
\psi ^{\left( k\right) }\left( z\right) & :=\frac{d^{k+1}}{dz^{k+1}}\log
\Gamma \left( z\right) =\frac{d^{k}}{dz^{k}}\left( \frac{\Gamma ^{\prime
}\left( z\right) }{\Gamma \left( z\right) }\right)   \notag \\
& =-\int_{0}^{1}\frac{\left[ \log (t)\right] ^{k}\ t^{z-1}}{1-t}dt,\ \left(
k\in \mathbb{N}_{0}\right)   \label{1.021A}
\end{align}%
and $\psi ^{\left( 0\right) }\left( z\right) =\psi \left( z\right) ,$
denotes the psi, or digamma function, defined by%
\begin{equation*}
\psi \left( z\right) =\frac{d}{dz}\log \Gamma \left( z\right) =\frac{\Gamma
^{\prime }\left( z\right) }{\Gamma \left( z\right) }.
\end{equation*}%
We also recall the relation, for $m=1,2,3,\ldots $%
\begin{equation}
H_{z-1}^{\left( m+1\right) }=\zeta \left( m+1\right) +\frac{\left( -1\right)
^{m}}{m!}\psi ^{\left( m\right) }\left( z\right) .  \label{1.021}
\end{equation}

The evaluation of series with harmonic numbers, or series which sum to
harmonic numbers dates back to the time of Euler. Since then many other
results have been obtained. The sum
\begin{equation*}
\sum_{k=1}^{n}\frac{\left( -1\right) ^{k+1}\binom{n}{k}}{k}=H_{n}^{\left(
1\right) }
\end{equation*}%
is well known and appears in a number of problems related to random
allocations and theory of records. Adamchik \cite{0.01} for example obtained
\begin{equation}
\sum_{k=1}^{n}\frac{H_{k}^{\left( 1\right) }}{k}=\left( H_{n}^{\left(
1\right) }\right) ^{2}+H_{n}^{\left( 1\right) }=2\sum_{r=1}^{n}\left(
-1\right) ^{r+1}\binom{n}{r}\frac{1}{r^{2}}.  \label{1.03}
\end{equation}%
Flajolet and Sedgewick \cite{2.1b} also got results of the type
\begin{equation}
6\sum_{k=1}^{n}\frac{\left( -1\right) ^{k+1}\binom{n}{k}}{k^{3}}=\left(
H_{n}^{\left( 1\right) }\right) ^{3}+3H_{n}^{\left( 1\right) }H_{n}^{\left(
2\right) }+2H_{n}^{\left( 3\right) }.  \label{FS}
\end{equation}%
There are many papers dealing with identities of harmonic numbers see, for
example,\ \cite{02}, \cite{13d}, \cite{sofsriv} and \cite{14a}. \

Binomial coefficients play an important role in many areas of mathematics,
including number theory, statistics and probability. The binomial
coefficient is defined as
\begin{equation*}
\binom{z}{w}=\frac{\Gamma \left( z+1\right) }{\Gamma \left( w+1\right)
\Gamma \left( z-w+1\right) }
\end{equation*}%
for $z$ and $w$ non-negative integers, where $\Gamma \left( x\right) $ is
the gamma function.

The representation of sums in closed form can in some cases be achieved
through a variety of different methods, including, integral representations,
transform techniques, Riordan arrays and the $W-Z$ method. The interested
reader is referred to the works of \cite{Choi}, \cite{ChoiSriv}, \cite{10b},
\cite{11b}, \cite{12b} and \cite{13b}.

The next lemma deals with the derivatives of binomial coefficients (see \cite%
{13c}).

\begin{lemma}
\label{lemma1} Let $j\geq 0,$ $n>0$ and let $Q(n,j)=\binom{n+j}{j}^{-1}$ be
an analytic function in $j.$ Then we have,%
\begin{equation*}
Q^{\left( 1\right) }(n,j)=\frac{dQ}{dj}=
\begin{cases}
-Q(n,j)P(n,j), & \left( j\in \mathbb{N}\right),  \\
&  \\
-H_{n}^{\left( 1\right) } & \left( j=0\right),
\end{cases}%
\end{equation*}%
where
\begin{equation*}
P(n,j):=\sum\limits_{r=1}^{n}\frac{1}{r+j}=\psi \left( j+1+n\right) -\psi
\left( j+1\right) .
\end{equation*}
\end{lemma}

We also have
\begin{align}\label{1.80}
Q^{\left( \lambda \right) }(n,j)&=\frac{d^{\lambda }Q}{dj^{\lambda }}\\
&=-\sum\limits_{\rho =0}^{\lambda -1}\binom{\lambda -1}{\rho }Q^{\left( \rho
\right) }(n,j)P^{\left( \lambda -1-\rho \right) }(n,j)\quad \left( \lambda
\in \mathbb{N}\backslash \left\{ 1\right\} \right) ,  \nonumber
\end{align}%
where $P^{\left( 0\right) }(n,j)=\sum_{r=1}^{n}\frac{1}{r+j}$ $\left( n\in
\mathbb{N}\right) ,$ and $Q^{\left( 0\right) }(n,j)=Q(n,j)$, and
\begin{align*}
P^{\left( i\right) }(n,j)& =\frac{d^{i}P}{dj^{i}}=\frac{d^{i}}{dj^{i}}\left(
\sum\limits_{r=1}^{n}\frac{1}{r+j}\right)  \\
& =(-1)^{i}i!\sum\limits_{r=1}^{n}\frac{1}{\left( r+j\right) ^{i+1}} \\
& =(-1)^{i}i!\left[ \zeta \left( i+1,j+1\right) -\zeta \left(
i+1,j+1+n\right) \right] \quad \left( i\in \mathbb{N}\right) ,
\end{align*}%
\begin{equation*}
\zeta \left( q,b\right) =\sum\limits_{k=0}^{\infty }\frac{1}{\left(
k+b\right) ^{q}}\quad \left( \func{Re}\left( q\right) >1;b\notin \mathbb{Z}%
_{0}^{-}\right)
\end{equation*}%
is the generalized (or Hurwitz) zeta function and $\mathbb{Z}_{0}^{-}$
denotes the set of non-positive integers.

Now we list some particular cases of Lemma \ref{lemma1},%
\begin{align*}
Q^{\left( 1\right) }(n,j)& =-\binom{n+j}{j}^{-1}\sum\limits_{r=1}^{n}\frac{1%
}{r+j}, \\
Q^{\left( 2\right) }(n,j)& =\binom{n+j}{j}^{-1}\left[ \left(
\sum\limits_{r=1}^{n}\frac{1}{r+j}\right) ^{2}+\sum\limits_{r=1}^{n}\frac{1}{%
\left( r+j\right) ^{2}}\right]  \\
& =\binom{n+j}{j}^{-1}\left[ \sum\limits_{r=1}^{n}\sum\limits_{s=1}^{r}\
\frac{2}{\left( r+j\right) \left( s+j\right) }\right] ,
\end{align*}%
\begin{multline*}
Q^{\left( 3\right) }(n,j)\\=-\binom{n+j}{j}^{-1}\left[ \left(
\sum\limits_{r=1}^{n}\frac{1}{r+j}\right) ^{3}+2\sum\limits_{r=1}^{n}\frac{1%
}{\left( r+j\right) ^{3}}+3\sum\limits_{r=1}^{n}\frac{1}{\left( r+j\right)
^{2}}\sum\limits_{r=1}^{n}\frac{1}{r+j}\right]
\end{multline*}%
and
\begin{multline*}
Q^{\left( 4\right) }(n,j)\\
=\binom{n+j}{j}^{-1}\left[ 6\sum\limits_{r=1}^{n}%
\frac{1}{\left( r+j\right) ^{2}}\left( \sum\limits_{r=1}^{n}\frac{1}{r+j}%
\right) ^{2}\right. +8\sum\limits_{r=1}^{n}\frac{1}{\left( r+j\right) ^{3}}%
\sum\limits_{r=1}^{n}\frac{1}{r+j} \\
+3\left( \sum\limits_{r=1}^{n}\frac{1}{\left( r+j\right) ^{2}}\right)
^{2}+\left. \left( \sum\limits_{r=1}^{n}\frac{1}{r+j}\right)
^{4}+6\sum\limits_{r=1}^{n}\frac{1}{\left( r+j\right) ^{4}}\right] .
\end{multline*}%
In the special case when $j=0$ we may write
\begin{equation}
Q^{\left( 1\right) }(n,0)=-H_{n}^{\left( 1\right) },  \label{A}
\end{equation}%
\begin{equation}
Q^{\left( 2\right) }(n,0)=\left( H_{n}^{\left( 1\right) }\right)
^{2}+H_{n}^{\left( 2\right) },  \label{B}
\end{equation}%
\begin{equation}
Q^{\left( 3\right) }(n,0)=-\left( H_{n}^{\left( 1\right) }\right)
^{3}-3H_{n}^{\left( 1\right) }H_{n}^{\left( 2\right) }-2H_{n}^{\left(
3\right) }\   \label{C}
\end{equation}%
and
\begin{equation}
Q^{\left( 4\right) }(n,0)=\left( H_{n}^{\left( 1\right) }\right)
^{4}+6\left( H_{n}^{\left( 1\right) }\right) ^{2}H_{n}^{\left( 2\right)
}+8H_{n}^{\left( 1\right) }H_{n}^{\left( 3\right) }+3\left( H_{n}^{\left(
2\right) }\right) ^{2}+6H_{n}^{\left( 4\right) }.  \label{D}
\end{equation}

Lemma \ref{lemma01} is restated and proved in the following theorem.

\section{First class of identities}

\begin{theorem}
\label{THM 1} Let $p$ be a positive integer. Then we get%
\begin{align}
\sum_{n=1}^{p}\frac{(-1)^{n+1}\binom{p}{n}H_{n}^{\left( 1\right) }}{n}
&=\int_{0}^{1}\frac{\log (1-x)}{x}\left( \left( 1-x\right) ^{p}-1\right) dx
\label{2.51} \\
&=H_{p}^{\left( 2\right) }.  \label{2.52}
\end{align}
\end{theorem}

\begin{proof}
We can write, for $t\in \mathbb{R},$
\begin{equation*}
\sum_{n=1}^{p}\frac{t^{n}\binom{p}{n}}{n~\binom{n+j}{j}}=\sum_{n=1}^{p}\frac{%
t^{n}\binom{p}{n}\Gamma \left( n\right) \Gamma \left( j+1\right) }{\Gamma
\left( n+j+1\right) }=\sum_{n=1}^{p}t^{n}\binom{p}{n}B\left( n,j+1\right)
\end{equation*}%
where $B\left( \cdot ,\cdot \right) $ is the classical beta function. Now
\begin{equation*}
\sum_{n=1}^{p}\frac{t^{n}\binom{p}{n}}{n~\binom{n+j}{j}}=\sum_{n=1}^{p}t^{n}%
\binom{p
}{n} B\left( n,j+1\right) =\sum_{n=1}^{p}t^{n}\binom{p%
}{n}\int_{0}^{1}x^{n-1}\left( 1-x\right) ^{j}dx.
\end{equation*}%
Differentiating the last resulting equation with respect to $j$ and putting $%
t=-1,$ we have%
\begin{equation*}
\sum_{n=1}^{p}\frac{\left( -1\right) ^{n+1}\binom{p}{n}}{n~\binom{n+j}{j}}%
\sum\limits_{r=1}^{n}\frac{1}{r+j}=\int_{0}^{1}\frac{\log (1-x)}{x}\left(
1-x\right) ^{j}\sum_{n=1}^{p}\left( -x\right) ^{n}\binom{p}{n}dx,\text{\ }
\end{equation*}%
which, upon setting $j=0$, yields%
\begin{align*}
\sum_{n=1}^{p}\frac{\left( -1\right) ^{n+1}\binom{p}{n}~H_{n}^{\left(
1\right) }}{n~}& =\int_{0}^{1}\frac{\log (1-x)}{x}\left( \left( 1-x\right)
^{p}-1\right) dx \\
& =\zeta \left( 2\right) -\psi ^{\prime }\left( p+1\right)  \\
& =H_{p}^{\left( 2\right) },
\end{align*}%
by the relation (\ref{1.021}).
\end{proof}

By recalling the inversion formula (see \cite{2.2}):
\begin{equation*}
g\left( n\right) =\sum\limits_{k}(-1)^{k}\binom{n}{k}f\left( k\right)
\Leftrightarrow f\left( n\right) =\sum\limits_{k}(-1)^{k}\binom{n}{k}g\left(
k\right) ,
\end{equation*}%
we find from the identity (\ref{2.52}):
\begin{equation*}
\sum_{n=1}^{p}\left( -1\right) ^{n+1}\binom{p}{n}~H_{n}^{\left( 2\right) }=%
\frac{~H_{p}^{\left( 1\right) }}{p~}.
\end{equation*}%
Theorem \ref{THM 1} can be generalized as follows.

\begin{theorem}
\label{THM 2} We have%
\begin{align*}
\sum_{n=1}^{p}\frac{(-1)^{n+1}\binom{p}{n}~Q^{\left( q\right) }\left(
n,0\right) }{n} &=\int_{0}^{1}\frac{\log ^{q}(1-x)}{x}\left( \left(
1-x\right) ^{p}-1\right) dx \\
&=\left( -1\right) ^{q+1}q!H_{p}^{\left( q+1\right) }\quad \left( p,q\in
\mathbb{N}\right)
\end{align*}%
and by inversion
\begin{equation*}
\sum_{n=1}^{p}(-1)^{n}\binom{p}{n}~H_{n}^{\left( q+1\right) }=\frac{%
(-1)^{q}~Q^{\left( q\right) }\left( p,0\right) }{p~q!}
\end{equation*}%
where $Q^{\left( q\right) }\left( \cdot ,\cdot \right) $ is defined by (\ref%
{1.80}).
\end{theorem}

\begin{proof}
The proof follows the same pattern as given in Theorem \ref{THM 1}, by
differentiating with respect to $j,$ $q$ times, and utilizing (\ref{1.021A}).
\end{proof}

\begin{example}
For $q=1$, we obtain the result (\ref{2.52}). For $q=4$ and using (\ref{D})
we get
\begin{displaymath}
\sum_{n=1}^{p}\frac{(-1)^{n}\binom{p}{n}\left( \left( H_{n}^{\left( 1\right)
}\right) ^{4}+6\left( H_{n}^{\left( 1\right) }\right) ^{2}H_{n}^{\left(
2\right) }+8H_{n}^{\left( 1\right) }H_{n}^{\left( 3\right) }+3\left(
H_{n}^{\left( 2\right) }\right) ^{2}+6H_{n}^{\left( 4\right) }\right) }{n} = 
24H_{p}^{\left( 5\right) }, 
\end{displaymath}
and
\begin{displaymath}
\sum_{n=1}^{p}(-1)^{n}\binom{p}{n}~H_{n}^{\left( 5\right) } \\
= \frac{\left( \left( H_{p}^{\left( 1\right) }\right) ^{4}+6\left(
H_{p}^{\left( 1\right) }\right) ^{2}H_{p}^{\left( 2\right) }+8H_{p}^{\left(
1\right) }H_{p}^{\left( 3\right) }+3\left( H_{p}^{\left( 2\right) }\right)
^{2}+6H_{p}^{\left( 4\right) }\right) }{24p}  .
\end{displaymath}
\end{example}

\section{Second class of infinite sum identities}

In this section we investigate another class of infinite sums involving
harmonic numbers. In particular we give closed form representations for the
class of sums $\sum_{n\geq 1}\frac{Q^{\left( q\right) }(n+b,0)}{n(n+b)}$.
The next lemma gives an alternative representation for $Q^{\left( q\right)
}(b,z).$

\begin{lemma}
\label{lemma 2} Let $b$ and $q$ be positive real integers and $z\in \mathbb{R%
}^{+}\cup \left\{ 0\right\} .$ Also let $Q^{\left( q\right) }(b,z)=\frac{%
d^{q}}{dz^{q}}=\binom{b+z}{z}^{-1}$ be an analytic function in $z$. Then we
have
\begin{align}
&\left( -1\right) ^{q-1}q!\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}%
\frac{r}{\left( r+z\right) ^{q+1}} \notag\\
&=Q^{\left( q\right) }(b,z)  \label{3.0}
\\
&=\frac{\left( -1\right) ^{q}q!~b}{\left( z+1\right) ^{q+1}}~_{\ \
q+2}F_{q+1}\left[ \left.
\begin{array}{c}
\overset{\left( q+1\right) -\text{terms}}{\overbrace{z+1,\ldots ,z+1}},1-b \\%
[5pt]
\underset{\left( q+1\right) -\text{terms}}{\underbrace{z+2,\ldots ,z+2}}%
\end{array}%
\right\vert 1\right] .  \label{3.01}
\end{align}
\end{lemma}

\begin{proof}
We first notice that
\begin{align*}
\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}\frac{1}{r+z}
&=\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}\int_{0}^{1}x^{r+z-1}dx \\
&=\int_{0}^{1}x^{z-1}\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}x^{r}~dx
\\
&=\int_{0}^{1}x^{z-1}\left( 1-x\right) ^{b}~dx=B\left( z,b+1\right)  \\
&=\frac{1}{z\binom{b+z}{b}}
\end{align*}%
and so
\begin{equation*}
z~\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}\frac{1}{r+z}=\frac{1}{%
\binom{b+z}{b}},
\end{equation*}%
which is a known result (see \cite{ChuYan} and \cite{2.2}). If we let
\begin{equation*}
F\left( b,z\right) =\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b}{r}\frac{1}{%
r+z},\text{ }
\end{equation*}%
then%
\begin{equation*}
zF\left( b,z\right) =Q\left( b,z\right) .
\end{equation*}%
Differentiating $q$-times with respect to $z,$ we have
\begin{equation*}
q~F^{\left( q-1\right) }\left( b,z\right) +z~F^{\left( q\right) }\left(
b,z\right)  =Q^{\left( q\right) }\left( b,z\right)
\end{equation*}%
or%
\begin{equation*}
\left( -1\right) ^{q-1}q!\sum_{r=0}^{b}\left( -1\right) ^{r}\binom{b
}{r} \frac{r}{\left( r+z\right) ^{q+1}}=Q^{\left( q\right) }(b,z).
\end{equation*}%
Chu \cite{02} has some special cases for $q=1.$ The hypergeometric function (%
\ref{3.01}) is defined by (\ref{1.02}). When $z=0,$ then
\begin{align}
\left( -1\right) ^{q-1}q!\sum_{r=1}^{b}\left( -1\right) ^{r}\binom{b
}{r}  \frac{1}{r^{q}}&=Q^{\left( q\right) }(b,0)  \label{3.1}
\\
&=\left( -1\right) ^{q}q!~b\,_{\ q+2}F_{q+1}\left[ \left.
\begin{array}{c}
\overset{\left( q+1\right) -\text{terms}}{\overbrace{1,\ldots ,1}},1-b \\[5pt]
\underset{\left( q+1\right) -\text{terms}}{\underbrace{2,\ldots ,2}}%
\end{array}%
\right\vert 1\right] ,\notag
\end{align}%
we note that (\ref{A}), (\ref{B}), (\ref{C}) and (\ref{D}) are special cases
of $Q^{\left( q\right) }(b,0)$ for $q=1,2,3,4.$
\end{proof}

The following two lemmas will be useful in the proof of the subsequent
theorem.

\begin{lemma}
\label{lemma 3} Let $b>0$ be a real positive number. Then we have
\begin{equation*}
-\frac{H_{b}}{b}=\int_{0}^{1}x^{b-1}~\log \left( 1-x\right) ~dx.
\end{equation*}
\end{lemma}

\begin{proof}
By symmetry%
\begin{align*}
\int_{0}^{1}x^{b-1}~\log \left( 1-x\right) ~dx& =\int_{0}^{1}\left(
1-x\right) ^{b-1}~\log x~dx \\
& =\int_{0}^{1}\sum_{r=0}^{b-1}\left( -1\right) ^{r}\binom{b-1}{r}x^{r}~\log
x~dx \\
& =-\sum_{r=0}^{b-1}\left( -1\right) ^{r}\binom{b-1}{r}\frac{1}{\left(
r+1\right) ^{2}} \\
& =\frac{1}{b}\sum_{r=1}^{b}\left( -1\right) ^{r}\binom{b}{r}\frac{1}{r}=-%
\frac{H_{b}}{b},\text{ }
\end{align*}%
where the last equality follows from (\ref{3.1}).
\end{proof}

\begin{lemma}
\label{lemma 4} Let $b$ be a positive real number. Then we have%
\begin{equation}
\sum_{n=1}^{\infty }\frac{H_{n+b}}{n\left( n+b\right) }=\frac{1}{b}Q^{\left(
2\right) }(b,0)=\frac{1}{b}\left( \left( H_{b}^{\left( 1\right) }\right)
^{2}+H_{b}^{\left( 2\right) }\right)   \label{6}
\end{equation}
\end{lemma}

\begin{proof}
From Lemma \ref{lemma 3}%
\begin{align*}
\sum_{n=1}^{\infty }\frac{H_{n+b}}{n\left( n+b\right) }&
=-\sum_{n=1}^{\infty }\frac{1}{n}\int_{0}^{1}\left( 1-x\right) ^{n+b-1}\log
x~dx, \\
& =\int_{0}^{1}x^{b-1}\log \left( 1-x\right) \left( -\sum_{n=1}^{\infty }%
\frac{x^{n}}{n}\right) ~dx \\
& =\int_{0}^{1}x^{b-1}\log ^{2}\left( 1-x\right) ~dx \\
& =\int_{0}^{1}\left( 1-x\right) ^{b-1}\log ^{2}x~dx \\
& =\int_{0}^{1}\sum_{r=0}^{b-1}\left( -1\right) ^{r}\binom{b-1
}{r}  x^{r}\log ^{2}x~dx \\
& =\sum_{r=0}^{b-1}\left( -1\right) ^{r}\binom{b-1
}{r}  \frac{2%
}{\left( r+1\right) ^{3}} \\
& =-\frac{2}{b}\sum_{r=1}^{b}\left( -1\right) ^{r}\binom{b
}{r}
\frac{1}{r^{2}} \\
& =\frac{1}{b}Q^{\left( 2\right) }(b,0)=\frac{1}{b}\left( \left(
H_{b}^{\left( 1\right) }\right) ^{2}+H_{b}^{\left( 2\right) }\right) ,
\end{align*}%
from Lemma~\ref{lemma 2} and (\ref{1.03}).
\end{proof}

The following theorem can now be stated.

\begin{theorem}
\label{THM 3} Let $b$ and $q$ be positive real integers. Then we have
\begin{align*}
\sum_{n=1}^{\infty }\frac{Q^{\left( q\right) }(n+b,0)}{n(n+b)}& =\frac{1}{b}%
Q^{\left( q+1\right) }(b,0) \\
& =\left( -1\right) ^{q-1}~q!_{\ \ q+2}F_{q+1}\left[ \left.
\begin{array}{c}
\overset{\left( q+1\right) -\text{terms}}{\overbrace{1,\ldots ,1}},1-b \\[5pt]
\underset{\left( q+1\right) -\text{terms}}{\underbrace{2,\ldots ,2}}%
\end{array}%
\right\vert 1\right]
\end{align*}%
where $Q^{\left( q\right) }(b,0)$ is defined by (\ref{3.1}).
\end{theorem}

\begin{proof}
By the induction principle, the case $q=1$ holds, from Lemma \ref{lemma 4}.
Now
\begin{align*}
\sum_{n=1}^{\infty }\frac{1}{n}\left( \frac{Q^{\left( q\right) }(n+b,0)}{%
(n+b)}\right) & =\sum_{n=1}^{\infty }\frac{1}{n}\int_{0}^{1}\left(
1-x\right) ^{n+b-1}\log ^{q}x~dx \\
& =\sum_{n=1}^{\infty }\frac{x^{n}}{n}\int_{0}^{1}x^{b-1}\log ^{q}\left(
1-x\right) ~dx,\text{ by symmetry,} \\
& =\int_{0}^{1}x^{b-1}\log ^{q+1}\left( 1-x\right) ~dx \\
& =\int_{0}^{1}(1-x)^{b-1}\log ^{q+1}x~dx \\
& =\int_{0}^{1}\sum_{r=0}^{b-1}\left( -1\right) ^{r}\binom{b-1}{r}x^{r}\log
^{q+1}x~dx \\
& =\left( -1\right) ^{q}\left( q+1\right) !\sum_{r=0}^{b-1}\left( -1\right)
^{r}\binom{b-1}{r}\frac{1}{\left( r+1\right) ^{q+2}},
\end{align*}%
and by Lemma \ref{lemma 2} we get
\begin{equation*}
\sum_{n=1}^{\infty }\frac{1}{n}\left( \frac{Q^{\left( q\right) }(n+b,0)}{%
(n+b)}\right) =\frac{\left( -1\right) ^{q}\left( q+1\right) !}{b}\sum_{r=1}^{b}\left(
-1\right) ^{r}\binom{b}{r}\frac{1}{r^{q+1}}=\frac{1}{b}Q^{\left( q+1\right)
}(b,0).
\end{equation*}%
This completes the proof of Theorem \ref{THM 3}.
\end{proof}

\begin{example}
For the case $q=2$
\begin{align*}
\sum_{n=1}^{\infty }\frac{\left( H_{n+b}^{\left( 1\right) }\right)
^{2}+H_{n+b}^{\left( 2\right) }}{n(n+b)}& =\frac{1}{b}\left( \left(
H_{b}^{\left( 1\right) }\right) ^{3}+3H_{b}^{\left( 1\right) }H_{b}^{\left(
2\right) }+2H_{b}^{\left( 3\right) }\right)  \\
& =\frac{2}{b}\sum_{r=1}^{b}\left( -1\right) ^{r+1}\binom{b}{r}\frac{1}{r^{3}%
}
\end{align*}%
this result also follows from (\ref{FS}). For $q=3$%
\begin{align*}
& \sum_{n=1}^{\infty }\frac{\left( H_{n+b}^{\left( 1\right) }\right)
^{3}+3H_{n+b}^{\left( 1\right) }H_{n+b}^{\left( 2\right) }+2H_{n+b}^{\left(
3\right) }}{n(n+b)} \\
& =\frac{1}{b}\left( \left( H_{b}^{\left( 1\right) }\right) ^{4}+6\left(
H_{b}^{\left( 1\right) }\right) ^{2}H_{b}^{\left( 2\right) }+8H_{b}^{\left(
1\right) }H_{b}^{\left( 3\right) }+3\left( H_{b}^{\left( 2\right) }\right)
^{2}+6H_{b}^{\left( 4\right) }\right)  \\
& =\frac{6}{b}\sum_{r=1}^{b}\left( -1\right) ^{r+1}\binom{b}{r}\frac{1}{r^{4}%
}=6_{\ 5}F_{4}\left[ \left.
\begin{array}{c}
1,1,1,1,1-b \\[5pt]
2,2,2,2%
\end{array}%
\right\vert 1\right] .
\end{align*}
\end{example}




\bigskip
\hrule
\bigskip

\noindent 2010 {\it Mathematics Subject Classification}:
Primary 05A10; Secondary 11B65, 05A19.

\noindent \emph{Keywords: } 
integral representation, binomial coefficient, identity, harmonic number.

\bigskip
\hrule
\bigskip

\vspace*{+.1in}
\noindent
Received  April 29 2012;
revised versions received   July 18 2012; August 31 2012.
Published in {\it Journal of Integer Sequences}, September 8 2012.

\bigskip
\hrule
\bigskip

\noindent
Return to
\htmladdnormallink{Journal of Integer Sequences home page}{http://www.cs.uwaterloo.ca/journals/JIS/}.
\vskip .1in

\end{document}

.
